Author's response to reviews

Title: Obesity: Physiologic Changes and Implications for Perioperative Management

Authors:

Vilma E Ortiz (vortiz@partners.org)
Jean Kwo (jkwo@mgh.harvard.edu)

Version: 2 Date: 11 December 2014

Author's response to reviews:

(1) A closer review of the original submission revealed that a reference was missing - line 247/248 needs reference

"The STOP-Bang questionnaire (Table 2) [76], developed specifically for use in surgical patients, has been validated in patients with a BMI > 30 [ref]"


This reference has been added. Of note, this is also the reference for Table 2.

(2) The author for reference #75 (line #667) was showing up wrong. It said "Apnea AToPMoPwOS". It should say “American Society of Anesthesiologists Task Force on Perioperative Management of Patients with Obstructive Sleep Apnea”

This has been corrected.